% Digest template for Intermag

\documentclass[journal,transmag]{IEEEtran}


\usepackage{cite}
\usepackage{graphicx}
\usepackage{amssymb,amsmath}




% correct bad hyphenation here
\hyphenation{mag-netic}


\begin{document}

% paper title
\title{Intermag Digest Preparation Instructions}



% author names and affiliations
% transmag papers use the long conference author name format.

\author{
	\IEEEauthorblockN{Erai Takahashi\IEEEauthorrefmark{1},~\IEEEmembership{Member,~IEEE}, Stan Laurel\IEEEauthorrefmark{2}, and Oliver N. Hardy\IEEEauthorrefmark{2},~\IEEEmembership{Senior Member,~IEEE}}
	\IEEEauthorblockA{\IEEEauthorrefmark{1}Institute of Magnetics, World Leading University, Monbetsu, 094-0011, Japan, taka@wlu.ac.jp}
	\IEEEauthorblockA{\IEEEauthorrefmark{2}Department of Physics, University of Life, Los Angeles, CA 90230, USA, stan@uol.edu}% <-this % stops an unwanted space
}



\IEEEtitleabstractindextext{%
\begin{abstract}
This template and instructions will help you to prepare a two-page digest for the Intermag conference. The general layout is almost identical to the standard template of the IEEE Transactions on Magnetics, except that Intermag digests are limited to two pages. You should use this template to prepare your contribution. The abstract should span the entire width of the page. If your abstract does not span the entire page width please use the latest version of the IEEEtran.cls file from the Intermag website. You may also use the MS Word template provided on the Intermag website. Accepted digests may be published on IEEE Xplore. 
\end{abstract}


\begin{IEEEkeywords}
About four key words or phrases in alphabetical order, separated by commas.
\end{IEEEkeywords}}


% make the title area
\maketitle

% no page numbering, or other headers / footers
\pagestyle{empty}
\thispagestyle{empty}

\IEEEpeerreviewmaketitle


\section{Introduction}
% The very first letter is a 2 line initial drop letter followed
% by the rest of the first word in caps.

\IEEEPARstart{T}{his document} is a template for Latex. The electronic version can be downloaded from the conference website.



\section{General Layout of the Two-Page Digest}

If you use this digest template you should end up with a two page, two column digest that meets the required specification. Please submit the resulting PDF to the conference.


\section{Useful Hints}
\subsection{Abbreviations and acronyms}

Define abbreviations and acronyms the first time they are used, even after they have been defined in the abstract. Commonly acceptable abbreviations such as IEEE, SI, MKS, ac, and dc do not have to be defined. Do not use abbreviations in the title unless they are unavoidable.


\subsection{Figures and tables}

Please use words rather than symbols to label the axes of graphs. As an example, write the quantity ``Magnetization'', or ``Magnetization, $M$'', not just ``$M$''. Put units in parentheses. Do not label axes only with units. As in Fig. \ref{MH_Loops}, for example, write ``Magnetization (emu/cm$^3$)'' or ``Magnetization (emu·cm$^{-3}$)'', not just ``emu/cm$^3$''. Do not label axes with a ratio of quantities and units. For example, write ``Temperature (K)'', not ``Temperature / K''. Multipliers can be confusing. Write ``Applied field (kOe)'' or ``Applied field (10$^3$ Oe)'', not ``Applied field (Oe $\times$ 1000)'' because the readers would not know whether the right axis label in Fig. \ref{MH_Loops} meant 20000 Oe or 0.02 Oe.


\begin{figure}
	\includegraphics[width=3.4in]{fig}
	\caption{Magnetization as a function of applied field.}
	\label{MH_Loops}
\end{figure}


Table \ref{My_Table} shows an example of a table. Note that there are no vertical lines in the table.



\begin{table}
\renewcommand{\arraystretch}{1.3}
\caption{Track pitch (TP), average switching probability $\langle P \rangle$, and user areal density (UAD) for dots with $d_1 = d_2 =$ 5 $\mathrm{nm}$.}
\centering
\begin{tabular}{l c c c c}
\hline
$T_{c1,2}$ (K) & $D$ (nm) & TP (nm) & $\langle P \rangle$ & UAD (Tbit/in$^2$)\\
\hline
460, 550 & 5 & 15.1 & 0.9620 & 6.55\\
460, 550 & 6 & 14.8 & 0.9867 & 7.83\\
460, 550 & 7 & 14.3 & 0.9947 & 8.59\\
\hline
530, 460 & 5 & 14.8 & 0.9464 & 6.09\\
530, 460 & 6 & 14.5 & 0.9800 & 7.64\\
530, 460 & 7 & 14.0 & 0.9933 & 8.68\\
\hline
\end{tabular}
\label{My_Table}
\end{table}



\subsection{Equations}

Number equations consecutively with equation numbers in parentheses flush with the right margin, as in \ref{My_Eq}. To make your equation more compact, you may use the solidus (/), the exp function, or appropriate exponents.  Use parentheses to avoid ambiguities in denominators. Punctuate equations when they are part of a sentence, as in

\begin{equation}
	\int_0^{r_2} F(r,\psi)dr d\psi = [ \sigma r_2 / (2 \mu_o)].
	\label{My_Eq}
\end{equation}

Be sure that the symbols in your equation have been defined before the equation appears or immediately following. Refer to “(\ref{My_Eq})”, not ``Eq. (\ref{My_Eq})'' or ``equation (\ref{My_Eq})'', except at the beginning of a sentence: ``Equation (\ref{My_Eq}) is …''. Please confine equations to one column width and break equations at appropriate algebraic symbols.


\subsection{Units}

Use either SI (MKS) or CGS as primary units. SI units are strongly recommended. Avoid combining SI and CGS units, such as current in Ampere and magnetic field in Oersted. If you must use mixed units, clearly state the units for each quantity in an equation.


\subsection{Conclusion}

A conclusion section is not required. Although a conclusion may review the main points of the paper, do not replicate the abstract in the conclusion. A conclusion might elaborate on the importance of the work or suggest applications and extensions.


\subsection{References}

Use bibtex for references \cite{Eason}. The sentence punctuation follows the brackets. In sentences, refer simply to the reference number, as in \cite{Box:RSM}. Do not use ``Ref. \cite{Houssameddine:STO}'' or ``reference \cite{Mallary:1Tb}'' except at the beginning of a sentence. Papers that have not been published should be cited as ``unpublished''. Papers that have been submitted for publication should be cited as ``submitted for publication''. Papers that have been accepted for publication but not yet specified for an issue should be cited as ``to be published''. Please give affiliations and address for private communications. Capitalize only the first word in a paper title, except for proper nouns and element symbols.

References should be used to place your work in context. Authors should not excessively cite their own prior publications. The editors of Magnetics Society journals and conference proceedings are opposed to all forms of bibliometric (impact factor) manipulation. Do not include gratuitous or irrelevant references to articles.


\subsection{Language}

The use of grammar and spelling checkers is strongly recommended. It is also suggested that you have the digest proofread by a native English-speaking colleague if your native language is not English.






% use section* for acknowledgement
%\section*{Acknowledgement}



\bibliographystyle{IEEEtran}
\bibliography{IEEEabrv,refs}







\end{document}

